Author's response to reviews

Title: Early postoperative cognitive dysfunction and postoperative delirium after anaesthesia with various hypnotics: study protocol for a randomised controlled trial - The PINOCCHIO trial -

Authors:

Federico Bilotta (bilotta@tiscali.it)
Andrea Doronzio (a.doronzio@libero.it)
Elisabetta Stazi (elisabetta.stazi@yahoo.it)
Luca Titi (luca.titi@hotmail.it)
Antonella Cianchi (antonellacianchi@libero.it)
Ivan Orlando Zeppa (jakie84@libero.it)
Giovanni Rosa (giov.rosa@libero.it)
Francesca Paola Paoloni (francesca.paoloni@yahoo.it)
Elisabeth Apollonia Abramowicz (Elisabeth.abramowicz@gmail.com)
Allison Spinelli (aspinell@montefiore.org)
Eugenia Ayrian (eugenia.ayrian@gmail.com)
Vladimir Zelman (vzelman@hsc.usc.edu)
Philip Lumb (lumb@usc.edu)
Sergio Bergese (sergio.bergese@osumc.edu)
Irene Asouhidou (iasouhidou@aol.com)
Polimnia Ioannou (pioan@gmail.com)
Ellise Delphin (edelphin@montefiore.org)

Version: 11 Date: 24 June 2011

Author's response to reviews: see over
Rome, 24/06/2011

Dear Dr. Goldsmith

Thank you for the peer-review feedback and the opportunity to resubmit this protocol entitled "Early postoperative cognitive dysfunction and postoperative delirium after anaesthesia with various hypnotics: study protocol for a randomised controlled trial - The PINOCCHIO trial". The feedback was extremely useful.

We have revised the manuscript according with your suggestion, modifying table format and typography.

Sincerely,

Federico Bilotta